Author's response to reviews

Title: HIV Surveillance in a Large, Community-Based Study: Results from the Pilot Study of Project Accept (HIV Prevention Trials Network 043)

Authors:

Estelle Piwowar-Manning (epiwowa@jhmi.edu)
Agnes Fiamma (afiamma@ucla.edu)
Oliver Laeyendecker (plaeyen1@jhmi.edu)
Michal Kulich (kulich@karlin.mff.cuni.cz)
Deborah Donnell (Deborah@scharp.org)
Greg Szekeres (GSzekeres@mednet.ucla.edu)
Laura Robins-Morris (lrobins@scharp.org)
Caroline Mullis (cmullis1@jhmi.edu)
Ana Vallari (ana.vallari@abbott.com)
John Hackett Jr (john.hackett@abbott.com)
Timothy Mastro (tmastro@fhi360.org)
Glenda Gray (gray@pixie.co.za)
Linda Richter (lrichter@hsrc.ac.za)
Michel Alexandre (michelwilner@yahoo.com)
Suwat Chariyalertsak (schariya@mail.med.cmu.ac.th)
Alfred Chingono (Alfred@uz-ucsf.co.zw)
Michael Sweat (sweatm@musc.edu)
Thomas Coates (tcoates@mednet.ucla.edu)
Susan H Eshleman (seshlem@jhmi.edu)

Version: 3 Date: 6 August 2011

Author's response to reviews:

Thank you for your review of our manuscript, “HIV Surveillance in a Large, Community-Based Study: Results from the Pilot Study of Project Accept (HIV Prevention Trials Network 043)” by Piwowar-Manning, et al. (manuscript: 1423810461544699). We have responded to each of the reviewers’ comments and have revised the manuscript to address issues raised in the review.

Please note that we have added two authors to the manuscript. One who participated in the testing of study samples, and one who assisted with the revisions and response to the reviewers’ comments.

We look forward to your review of the revised manuscript, and hope that you find the revised manuscript acceptable for publication.